Author’s response to reviews

Title: Annual decline in forced expiratory volume and airway inflammatory cells and mediators in a general population-based sample

Authors:

Natalia Kononova (natalia.kononova@medisin.uio.no; nataliapromenaden@gmail.com)

Liv Ingunn Bjoner Sikkeland (l.i.b.sikkeland@medisin.uio.no)

Faiza Mahmood (Faiza.Mahmood@ahus.no)

Maria Vistnes (maria.vistnes@medisin.uio.no)

Johny Kongerud (johny.kongerud@medisin.uio.no)

Gunnar Einvik (gunnar.einvik@medisin.uio.no)

Vidar Søyseth (vidar.soyseth@medisin.uio.no)

Version: 5 Date: 22 Nov 2018

Author’s response to reviews:

Cover letter

Editor Comments (EC):

EC1: “Thank you for providing an Ethics and consent for participation statement for your Declarations section, please list the full, unabbreviated name of the ethics review board that approved of your study.”

Author reply: “now it corrected.” See page 23 line 6.

EC2: ”Please place the titles/captions of all figures at the end of the main manuscript after the References section under a newly created “Figure legend” heading.”

Author reply:” now it added at page 28, line 19,20”.
EC3:” Please upload English language versions of all questionnaire/s developed specifically for use in this study as supplementary files. If the questionnaires are not yours or are already published by you, please supply references and/or links to them. Please add a “Supplementary files” section where you list the following information about your questionnaire/s:

- File name
- Title of data
- Description of data”

Author reply:” now it added at page 28, line 21-27”.